Author's response to reviews

Title: The Association Between Celiac Disease and Eosinophilic Esophagitis in Children and Adults

Authors:

Michael J Stewart (michael.stewart@dal.ca)
Eldon Shaffer (shaffer@ucalgary.ca)
Stephan J Urbanski (stefan.urbanski@cls.ab.ca)
Paul L Beck (plbeck@ucalgary.ca)
Martin A Storr (gidoc@gmx.com)

Version: 3 Date: 21 May 2013

Author's response to reviews: see over
Subject: The Association Between Celiac Disease and Eosinophilic Esophagitis in Children and Adults; Revised manuscript

Dear Dr. Foote,

We would like to thank you as well as the Reviewers for their valuable input and are pleased that you have recommended our manuscript for publication in BMC Gastroenterology.

Thank you also for forwarding the comments from the third Reviewer – even given the confusion. We do value the comments and feel that the previous revisions have addressed most of the concerns. We have made some additional minor changes to the manuscript in order to further address issues raised by Reviewer 3.

This revised manuscript uses red font to indicate revisions from the previous version.

Best Regards,

Michael Stewart and Martin Storr